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Abstract

Viscous fingering is an interfacial instability that occurs when multiple fluids displace each
other. This research focuses on the interface instability during immiscible displacement
of shear-thinning fluids by CO2. By controlling velocity and applying heat to the upper
and lower walls, the influence of velocity and temperature on viscous fingering during
CO2 displacement is investigated. Moreover, by modifying the geometric conditions of
the classical Hele-Shaw cells (HSCs), a novel analytical framework for viscous fingering
is proposed. The primary methodology involves implementing a minute depth gradient
distribution within the HSC, coupled with the Volume of Fluid (VOF) multiphase model,
which systematically reveals the dynamic suppression mechanism of shear-thinning effects
on viscous finger bifurcation. The results indicate that temperature elevation leads to
increased sweep efficiency, reduced residual non-Newtonian fluid in the displaced zone,
and enhanced displacement efficiency. Furthermore, increased velocity leads to reduced
sweep efficiency. However, at lower velocities, displacement efficiency remains relatively
low due to limited sweep coverage. The direction and magnitude of the depth gradient
significantly govern the morphology and extension length of viscous fingering. Both
positive and negative depth gradients promote fingering development on their respective
sides, as the gradient establishes anisotropic permeability that prioritizes flow pathways in
specific orientations, thereby intensifying finger propagation.

Keywords: viscous fingering; non-Newtonian fluid; carbon dioxide; interface control;
CFD; VOF

1. Introduction
Concerning the Hele-Shaw (two closely spaced parallel plates), observations have

shown that during displacement processes like water flooding to displace oil, the less
viscous displacing fluid forces the more viscous crude oil to move, forming dendritic
intrusion channels. This diversion effect, resulting from viscosity mismatch, is collectively
referred to as viscous fingering [1–3]. This effect manifests extensively across natural
and engineered systems, including groundwater infiltration in subsurface aquifers [4],
reservoir displacement processes in petroleum production [5], curtain coating defects in
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industrial applications [6], and transdermal drug diffusion in biomedical contexts [7]. The
dual-edged nature of this phenomenon has garnered cross-disciplinary attention. For
beneficial applications in materials design, it enables precise modulation of dendritic fractal
architectures [8]—facilitating the fabrication of catalyst supports with high specific surface
area in nanoporous materials [9], structural coloration surfaces for bio-inspired functional
coatings [10], and materials endowed with unique photonic/electronic responsiveness in
microfluidic devices [11]. Conversely, for negative suppression in petroleum extraction,
viscous fingering induces premature breakthrough of displacing fluids, reducing crude
oil recovery rates by 10–30% [12–14]; in microchemical systems, this phenomenon induces
reactant mixing inhomogeneity and microchannel clogging, effects that become markedly
accentuated in shear-thinning non-Newtonian fluids—such as polymer solutions and
blood—within hemodynamic or rheological contexts [15–17]. Consequently, achieving
active control over interfacial stability across diverse scenarios—such as fostering beneficial
fingering in materials synthesis while suppressing detrimental instabilities in petroleum
extraction—emerges as a pivotal frontier for enhancing technical efficacy.

A substantial body of international scholarship has conducted multifaceted in-depth
investigations into the viscous fingering phenomenon. Chuoke et al. derived the critical
velocity for viscous fingering initiation from planar interface instability using perturbation
theory [18]. Subsequently, numerous researchers have conducted extensive theoretical and
experimental investigations centered on this foundational framework. S. A. Abdul Hamid
et al. investigated the fingering phenomenon during invading water penetration into polymer
slugs in secondary polymer flooding through a coupled approach integrating numerical
simulations and analytical methods [19]. This study establishes an approximate method for
predicting the dynamic process of polymer slug volume growth caused by water fingering
under no-adsorption conditions. This method derives a concise analytical expression that can
be used to estimate the minimum slug size required to prevent water fingering from compro-
mising slug integrity before polymer breakthrough. Alan et al. used numerical simulation to
predict the incremental oil recovery and water cut reduction characteristics caused by polymer
injection [20]. Furthermore, the physical mechanism of this phenomenon has been revealed
and a predictive framework established through the latest modeling approach. Salmo et al.
verified the novel immiscible viscous fingering modeling approach through experimental
studies [21]. The experiment obtained spatiotemporal images of two-dimensional immisci-
ble fingering evolution through X-ray scanning technique, convincingly demonstrating the
effectiveness of the novel approach. As a core parameter in unsteady-state immiscible displace-
ment simulation, the fractional flow curve ( fw) function exhibits systematic variations closely
related to the viscosity ratio ( µ0/µw), with such variations possessing well-defined physical
significance. Currently, numerous researchers have conducted extensive experimental and
theoretical studies focusing on Newtonian displaced fluids [22–26], investigating the effects of
various parameters—including fluid properties, flow inertia, channel geometry, permeability,
and capillary pressure—on viscous fingering.

Compared with Newtonian fluids, non-Newtonian fluids flowing in HSCs have re-
ceived relatively less attention. However, it is noteworthy that fluid rheology plays a
pivotal role in the evolution of viscous fingers. Viscous fingering is a flow instability caused
by viscosity contrast between two liquids, where even minute perturbations can induce sig-
nificant alterations in the motion of the contact line. Yang et al. constructed a pore network
model to simulate non-Newtonian fluid two-phase flow [27], and identified a transition
mechanism from capillary fingering to viscous fingering during the displacement process.
For Ellis fluids, the stronger the shear-thinning effect, the more readily the injected fluid
invades small-radius pore throats, thereby suppressing viscous fingering development,
enhancing displacement efficiency, and significantly reducing the scale of the transition
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zone. Ahmadikhamsi et al. employed a radial HSC experimental setup to investigate
viscous fingering phenomena in non-Newtonian fluids induced by the displacement of oil
phases using Carbopol®940 polymer solutions (with/without SDS surfactant) [28]. The
experiments revealed that, contrary to conventional Saffman–Taylor instability charac-
teristics, the injection rate of polymer solutions significantly affects the morphological
evolution of viscous fingers. The study demonstrates that surfactant addition not only
alters system properties by reducing interfacial tension, but also decreases solution vis-
cosity, thereby collectively enhancing capillary number effects. Shokri et al. investigated
thermal viscous fingering instabilities during the displacement of Newtonian fluids by
shear-thinning non-Newtonian fluids in heterogeneous porous media [29]. The results
indicate that a decrease in the thermal retardation coefficient intensifies flow instabilities,
numerical simulations quantitatively characterized perturbation mechanisms of stratified
heterogeneity on the coupled evolution of temperature and concentration fields, identi-
fying switching thresholds for fingering patterns under synergistic effects of heat-mass
transfer and rheological properties. Logvinov et al. discovered that the fluid diffusion
rate exhibits strong dependence on the power law index n [30]. Mafei et al. elucidated the
regulatory mechanism of depth gradients on interfacial stability by introducing minute
depth gradients or modifying the geometric configuration of the HSC top plate, thereby
validating the efficacy of depth gradient manipulation in controlling viscous fingering
behavior during air-driven displacement of non-Newtonian fluids [31]. In studies focusing
on the displacement of non-Newtonian fluids, the influence of factors such as the properties
of the non-Newtonian fluid, the HSC configuration, and the pore structure on the viscous
fingering phenomenon was investigated. While displacing phases such as water and air
are predominantly selected as conventional media in the studies, research specifically
addressing the displacement of non-Newtonian fluids by CO2 has rarely been explored.

In summary, this study will utilize numerical simulation methods to investigate and
optimize the CO2 displacement shear-thinning fluid process within a three-dimensional
HSC. The temperature of the upper and lower walls and the CO2 displacement velocity will
be used as variables to elucidate the effects on viscous fingering in the three-dimensional
model. In addition, the HSC will be geometrically modified by introducing varying depth
gradients to modulate the fingering behavior.

2. Numerical Method
In an HSC system, the governing equations are typically formulated using the depth-

averaged Darcy’s law. This implies that the depth-averaged velocity of each fluid phase
along the depth direction can be expressed by Equation (1).

uj = − h2

12µj
∇pj (1)

where pj and uj are the phase j depth-averaged pressure and velocity vectors, respectively.
Due to the overall low level of fluid motion velocity, the flow state can be considered

incompressible. Consequently, the continuity equation can be transformed into:

∇·uj = 0 (2)

where u represents the velocity vector. The corresponding momentum equation (or equation
of motion) can be expressed as:

ρ

[
∂u
∂t

+ (u·∇)u
]
= ∇·

(
−pI + µ

[
∇u + (∇u)T

])
+ Fst (3)
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In the formula, p represents pressure, ρ represents density, µ represents dynamic
viscosity, I represents unit vector, t represents time, and Fst represents surface tension.

In this study, ANSYS Fluent 2021 R1 (commercial software) is utilized to conduct
the numerical simulations. To capture the two-phase interface and characterize the
morphological features of viscous fingering, the VOF method, based on a fixed Eulerian
computational grid, is employed to track the free boundary. This method assumes that
the fluids involved in the advected fluxes are immiscible and defined within the computa-
tional cells. Furthermore, the VOF method adopts a single coupled pressure equation and
a single momentum equation system for each dimension. For multiphase flow modeling,
this method formulates a shared velocity field based on the mass-weighted averaged
mass and momentum transport equations. These equations, neglecting interphase mass
transfer, can be defined as:

∂

∂t
(rαρα) +∇·

(
rαρα

→
Uα

)
= 0 (4)

∂
∂t

(
rαρα

→
Uα

)
+∇·

(
rαρα

→
Uα
⊗ →

Uα

)
= ∇·

(
rαµα

(
∇

→
Uα +

(
∇

→
Uα

)T
))

−rα∇p + rαρα
→
g +

→
Fst

(5)

where the subscript α refers to the gas (g) and liquid (l) phases such that rl + rg = 1. In

addition,
→

Uα represents the velocity field of phase α, rα—volume fraction, ρα—density,

p—pressure, µα—viscosity,
→
g —gravity and

→
Fst represents the surface tension.

In this study, the parameters of the underlying 3D rectangular HSC are as follows:
the length is 100 mm, the width is 20 mm, and the gap (depth, d) between the upper and
lower plates is 1 mm. The specific configuration is illustrated in Figure 1. Furthermore,
a series of variations in its depth gradient were conducted. During the entire simulation
process, high-quality structured grids were employed, with the total number of cells
approximately 1.4 million. Furthermore, to enhance simulation accuracy, mesh refinement
was performed in the boundary layer region (as shown in Figure 2b). When there were more
than 1.4 million elements in the meshed computational domain, the grid independent test,
as shown in Figure 2b, revealed that the velocity at the out-flow did not depend on grid size.
As a result, we took into account the 1,432,856 element count in the computational, domain.

Figure 1. Three-dimensional model of the HSC.

In terms of boundary condition settings, CO2 displaces the non-Newtonian fluid at
a constant velocity of 0.02 m/s which is uniformly distributed at inlet, while the outlet
adopts an ambient atmospheric average static pressure outlet condition, where the relative
pressure Prel is 0 Pa. In addition, the entire displacement time is 3.5 s. In the calculation
settings, a second-order upwind scheme is adopted to balance accuracy and stability,
avoiding excessive dissipation of the first-order scheme or oscillation of the QUICK scheme
under strong gradients. For pressure velocity coupling, the PISO algorithm is used, which
converges faster than the SIMPLE algorithm and is particularly suitable for transient
situations. To solve the interface smearing problem in the VOF method, the time step
is 0.002 s, ensuring that the interface movement does not exceed 1 or 2 grid cells within
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each time step. The displaced phase is a non-Newtonian fluid, with primary consideration
given to the impact of its shear-thinning behavior on the viscous fingering phenomenon.
During the numerical simulation process, a power-law non-Newtonian fluid model was
implemented in the calculations.

Figure 2. Grids of the computational domain (a) and verification of mesh independence (b).

µ = µ0γn−1 (6)

where γ is the shear rate, and n is the power law exponent. The fluid is classified into
shear thickening for n > 1, shear thinning for n < 1, and the fluid recovers the Newtonian
behavior at n = 1. For shear thinning fluids, the smaller the value of n, the more significant
the shear thinning effect. In this study, we consider medium shear thinning fluid where
n = 0.4 as the research object.

3. The Effects of Wall Temperature on Viscous Fingering During
Displacement Processes

In the process of CO2 displacing non-Newtonian fluids, to investigate the underlying
mechanism of temperature on the viscous fingering phenomenon, three temperature con-
ditions, T1 = 300 K, T2 = 305 K, and T3 = 320 K, were selected for simulation studies. The
computational domain and boundary conditions are illustrated in Figure 3. At the initial
moment, the entire fluid domain is occupied by the non-Newtonian fluid (Fluid 1), which
is maintained at an ambient temperature of 298.15 K; subsequently, ambient-temperature
(298.15 K) CO2 (Fluid 2) begins to enter the domain. In the numerical setup, the upper
and lower walls were configured with no-slip boundary conditions, which is the same pro-
cessing method with Singh [32]. Their temperatures were actively maintained at specified
values T1, T2, and T3 respectively to achieve continuous fluid heating.

Figure 3. Computational model configuration implementing wall heating technology (with tempera-
ture control points).

During the computational process, the energy equation must be solved due to the
involvement of thermal diffusion and convection. The viscosity of the non-Newtonian fluid
was redefined by selecting the ‘shear rate and temperature-dependent’ model. Under this
model, the viscosity is governed by the constitutive equation given in Equation (8). It can
be observed that viscosity exhibits an inverse relationship with temperature, specifically, as
temperature increases, viscosity tends to decline.

µ = η
( .
γ
)

H(T) (7)
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where H(T) is the temperature dependence, known as the Arrhenius law.

H(T) = exp
[

α

(
1

T − T0
− 1

Tα − T0

)]
(8)

where α is the ratio of the actication energy to the thermodynamic constant and Tα is a
reference temperature for which H(T) = 1. T0, which is the temperature shift, is set to 0 by
default and corresponds to the lowest temperature that is thermodynamically acceptable.
Therefore Tα and T0 are absolute temperatures.

From Figure 4, it is distinctly observed that regardless of temperature magnitude,
symmetric fingering structures develop in the CO2 displacement front at t = 1 s, with
cavitation bubble nuclei emerging near the fingering roots. However, with increasing
temperature, the fingering propagation distance decreases, the fingering root thickens, and
the intermediate cavitation zone progressively contracts. At t = 1.5 s in the displacement
process, under the 300 K and 305 K conditions, degenerative thinning occurs in the upper
fingering structure. In contrast, at 320 K, both fingering branches develop well-maintained
profiles with enhanced interfacial stability between the displacing and displaced phases.
From Figure 5, it is quantitatively determined that at t = 1 s with a temperature of
300 K, the fingering propagation distance is measured as 0.043 m. When the temperature is
increased to 320 K, this distance shortens to 0.035 m. Similarly, at t = 1.5 s under 300 K,
the inferior fingering branch propagates to 0.076 m, whereas at 320 K the superior fingering
extension reaches 0.064 m. This quantitatively confirms that elevated temperature reduces
the maximum fingering penetration distance by approximately 19.8%, corresponding to
enhanced stabilization of displacement fronts.

Figure 4. Morphological contours of viscous fingering at t = 1 s (left) and t = 1.5 s (right) under
distinct temperature conditions.

Figure 6 defines sweep efficiency φ: the ratio of the CO2-swept area Av (area enclosed
by the black contour line) to the reference area A0 (green-shaded area demarcated by the
maximum penetration front of fingering structures).

φ = Av/A0 (9)

φ increases with rising temperature. This indicates a reduction in residual non-
Newtonian fluid within the displaced zone and a corresponding enhancement in dis-
placement efficiency. Due to the increase in temperature, the kinetic energy of molecular
thermal motion is enhanced, breaking the existing weak interactions, resulting in a macro-
scopic decrease in viscosity, which helps CO2 to be better absorbed in the reservoir. In
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addition, an in-crease in temperature will accelerate the diffusion of molecules at the in-
terface, making it easier for displacement phase molecules to diffuse into the displaced
phase, expanding the displacement range and improving displacement efficiency. This
holds significant implications for enhancing hydrocarbon recovery in oil and gas reservoirs,
as expanded sweep coverage enables the mobilization and displacement of previously
immobile fluid phases trapped in pore networks.

Figure 5. Spatial–temporal evolution of viscous fingering front positions under varying temperature
conditions at distinct time instants.

Figure 6. Quantitative mapping of sweep region and corresponding sweep efficiency φ evolution
during CO2 displacement with increasing temperature.

It can be concluded that at both t = 1 s and t = 1.5 s, the sweep efficiency

4. Effects of CO2 Displacement Velocity on Viscous Fingering Instabilities
To investigate the effects of CO2 displacement velocity on viscous fingering, three repre-

sentative velocity magnitudes, 0.01 m/s, 0.02 m/s, and 0.04 m/s were selected for systematic
analysis. From Figure 7, the viscous fingering patterns at a unified time instant of t = 0.8 s
are analyzed. At the displacement velocity of 0.01 m/s, the phase interface exhibits relatively
regular and smoothed morphology, with a concentrated distribution of air volume fraction,
demonstrating stable advancement of the displacement front. This stability phenomenon due
to under low-velocity conditions, the fluid interaction between CO2 and shear-thinning fluids
exhibits relatively weak coupling. Consequently, viscous forces dominate the flow dynamics,
effectively constraining interfacial deformation and suppressing diffusive mixing between
the immiscible phases. When the displacement velocity increases to 0.02 m/s, the interfa-
cial morphology becomes significantly more complex, manifesting dichotomous symmetric
fingering branches. Cavitation inception emerges near the fingering roots, accompanied by
an expanded spatial distribution of the air volume fraction field. This phenomenon origi-
nates from increased inertial forces at higher velocities, which progressively overcome the
viscous resistance of the displaced fluid. Consequently, CO2 more readily penetrates the
non-Newtonian phase, disrupting the interfacial stability maintained at low velocities and
triggering nonlinear amplification of fingering instabilities. Under the high-velocity condition
of 0.04 m/s, the displacement interface transitions into a turbulent-like state. The inferior fin-
gering branch exhibits near-complete detachment from its route structure, while the air volume
fraction distribution becomes spatially expanded and fragmented across broader regions of
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the flow domain. This intensified flow behavior arises because inertial forces substantially
dominate over viscous resistance at elevated velocities. Consequently, CO2 rapidly penetrates
and convectively disperses within the shear-thinning fluid matrix, culminating in complex
displacement morphologies characterized by turbulent-like interfacial dynamics. Furthermore,
with increasing displacement velocity, the fingering penetration distance exhibits significant
augmentation. At 0.04 m/s, the maximum finger extension reaches 0.078 m, exceeding the
value observed at 0.01 m/s by a factor of approximately 9.75 (as quantified in Figure 8). At
elevated velocities, despite achieving greater penetration distances, the dispersed flow patterns
of CO2 result in extensive spatial bypassing of the shear-thinning fluid. This leads to height-
ened entrapment of residual non-Newtonian phase within un-swept pore networks, ultimately
diminishing macroscopic displacement efficiency. Consequently, the sweep efficiency φ under
varying velocity conditions was analyzed (as validated in Figure 9). A monotonic decline in
φ is observed with increasing velocity magnitude. At the velocity of 0.01 m/s, φ peaks at
83%, representing the optimal displacement coverage scenario. However, at lower velocities,
the constrained sweep coverage combined with sluggish advancement of the displacement
front results in diminished volumetric displacement rates per unit time, leading to suboptimal
displacement efficiency on a temporal basis. As displacement velocity increases, the sweep cov-
erage expands and the displacement front advances at accelerated rates. This synergistic effect
substantially enhances the volumetric displacement flux per unit time, leading to measurable
improvements in displacement efficiency. Despite the expanded spatial coverage observed
at elevated velocities, runaway fingering instabilities induce severe flow channeling. This
results in heterogeneous displacement patterns characterized by diminished sweep efficiency
φ and elevated retention of residual non-Newtonian fluid within unswept zones. Therefore, in
diverse engineering applications, a holistic optimization approach must be adopted to balance
sweep efficiency (φ) against displacement efficiency. This necessitates strategic selection of
operational parameters—particularly displacement velocity and temperature—to maximize
hydrocarbon recovery while minimizing residual trapping.

Figure 7. Morphological contours of viscous fingering patterns under distinct velocity magnitudes at
t = 0.8 s.
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Figure 8. Spatiotemporal evolution and quantitative comparison of viscous fingering front positions
across distinct velocity magnitudes (0.01 m/s, 0.02 m/s, 0.04 m/s) at t = 0.8 s.

 

Figure 9. Quantitative comparison of sweep efficiency (φ) across distinct velocity magnitudes
(0.01 m/s, 0.02 m/s, 0.04 m/s) at t = 0.8 s.

5. Effects of Depth Gradient on Viscous Fingering Instabilities in
Modified HSC

It is well-established that once the physical properties of fluids are fixed, the parameter
space available for flow manipulation becomes severely constrained. To overcome this limita-
tion, a microscale depth gradient was incorporated into the HSC configuration—providing a
novel geometric dimension to modulate classical viscous fingering dynamics. In this modified
configuration, the HSC adopts a tapered geometric profile along the direction of fluid dis-
placement. The investigation encompasses two distinct tapered configurations: positive and
negative depth gradients (Figure 10). Both configurations exhibit a constant depth gradient
magnitude along the primary flow direction.

Figure 10. Introducing depth gradient in an HSC: positive gradient (left) and negative gradient (right).

The flow dynamics at t = 1 s were analyzed across distinct depth gradient configura-
tions. From Figure 11, it is observed that in the absence of a depth gradient, the displace-
ment of the shear-thinning fluid manifests dichotomous symmetric fingering branches.
The interfacial progression exhibits relative uniformity with a stabilized displacement
front, while cavitation inception emerges proximal to the bifurcation zone of fingering
roots. Quantitatively, this configuration yields the shortest penetration distance (0.078 m at
t = 1 s), as rigorously quantified in Figure 12. Under positive depth gradient conditions, the
displacement of the shear-thinning fluid generates asymmetric fingering structures between
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the upper and lower domains. The superior fingering branch exhibits enhanced development
compared to its inferior counterpart, with moderate cavitation confined to micro-scale nuclei
near the fingering roots. Furthermore, as the positive depth gradient intensifies from 0.5◦

to 1.0◦, the superior fingering branch maintains morphological stability without significant
alterations, while the inferior branch undergoes progressive degeneration. Conversely, un-
der negative depth gradient conditions, the inferior fingering branch demonstrates superior
development relative to its superior counterpart, accompanied by pronounced cavitation
phenomena localized at the fingering roots. Through quantitative comparison in Figure 12,
it is demonstrated that as the depth gradient intensifies from −0.5◦ to −1.0◦, the fingering
propagation distance exhibits a monotonic increase. At the depth gradient of −1.0◦, viscous
fingering achieves its maximum penetration length of approximately 0.053 m, representing
the farthest displacement extent observed in the study. This phenomenon arises from the flow
constriction induced by the negative depth gradient, which hydrodynamically focuses CO2

into a dominant fingering channel. This confinement generates a high-speed narrow band
flow, resulting in localized velocity amplification—as quantitatively visualized in Figure 13.
In contrast, the fingering propagation distance under positive depth gradient conditions is
less than that achieved with negative gradients yet exceeds the displacement extent observed
in the absence of any depth gradient. Consequently, both the direction and magnitude of the
depth gradient exert profound influences on viscous fingering morphology and extension dy-
namics, as governed by the following mechanistic principles, both positive and negative depth
gradients selectively enhance the development of laterally preferential fingering branches,
whereas the absence of depth gradient yields a symmetric dual-branch pattern with balanced
interfacial progression. This phenomenon fundamentally stems from alterations in flow field
distribution induced by depth variations along the flow path. The presence of a depth gradient
establishes preferential flow pathways in specific spatial orientations, thereby amplifying
fingering instabilities through localized flow intensification.

Figure 11. Contour maps of fluid phase distribution for shear-thinning fluids under distinct depth
gradient configurations.
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Figure 12. Quantitative comparison of viscous fingering front positions across distinct depth gradients
at t = 1 s.

Figure 13. Visualization of velocity field distributions under distinct depth gradient configurations at
t = 1 s.

From the velocity contour maps in Figure 13, it is quantitatively evident that asymmet-
ric velocity distributions persist between dual fingering branches across all depth gradient
configurations, including the baseline case with zero gradient. Particularly under depth
gradient conditions, the velocity magnitude within degenerated fingering branches ap-
proaches quasi-stagnant flow regimes. From Figure 14, it is quantitatively demonstrated
that as the depth gradient transitions from negative to positive values (−1◦ to +1◦), the
maximum velocity within fingering structures undergoes progressive attenuation. This
reduction stems from the geometric transition of the flow conduit from convergent (nar-
rowing) to divergent (widening) profiles, which fundamentally decelerates flow velocities
through hydrodynamic expansion effects. Although the flow velocity under positive depth
gradient conditions is minimized, the geometric expansion of the flow conduit significantly
reduces viscous resistance. This facilitates focused advancement of CO2 along the dominant
fingering channel, ultimately achieving greater penetration distance compared to scenarios
devoid of depth gradients. Therefore, depth gradients fundamentally modulate viscous
fingering morphology and propagation dynamics by altering the geometric confinement
of flow conduits, thereby inducing strategic redistribution of velocity fields. In engineer-
ing applications, deliberate modulation of depth gradients within flow conduits enables
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strategic control over displacement behaviors, thereby facilitating targeted suppression or
triggering of interfacial instabilities as required by operational objectives.

Figure 14. Quantitative analysis of fingering penetration distance and maximum intra-fingering
velocity under distinct depth gradients.

6. Conclusions
In this study, a numerical simulation methodology was employed to investigate the

process of CO2 displacing shear-thinning fluids within a three-dimensional HSC. Distinct
investigations were systematically conducted to quantify the effects of velocity magni-
tude and temperature conditions on the displacement dynamics. Furthermore, to achieve
targeted modulation of viscous fingering processes, the three-dimensional HSC was en-
gineered with modified geometric configurations incorporating variable depth gradients.
The principal findings are summarized as follows:

(1) Through comparative analysis of displacement behaviors at 300 K, 305 K, and 320 K,
it is consistently observed that symmetrically bifurcated fingering structures develop
across all temperature conditions, accompanied by cavitation nucleation phenomena
localized at the fingering roots. However, with increasing temperature, the fingering
propagation distance decreases, the root thickness of the fingering structures increases,
and the intermediate cavitation zone progressively contracts. Furthermore, the sweep
efficiency φ exhibits a positive correlation with increasing temperature, indicating a
reduction in residual non-Newtonian fluid within the displaced zones and a conse-
quent enhancement in displacement efficiency. Due to the increase in temperature, the
kinetic energy of molecular thermal motion is enhanced, breaking the existing weak
interactions, resulting in a macroscopic decrease in viscosity, which helps CO2 to be
better absorbed in the reservoir. In addition, an increase in temperature will accelerate
the diffusion of molecules at the interface, making it easier for displacement phase
molecules to diffuse into the displaced phase, expanding the displacement range
and improving displacement efficiency. This holds profound strategic significance
for enhancing hydrocarbon recovery in both oil and gas reservoirs. The thermally
expanded sweep coverage enables efficient mobilization and displacement of previ-
ously inaccessible fluid phases trapped in complex pore networks, thereby unlocking
substantial volumes of challenging reserves.

(2) To investigate the influence of velocity on fingering, three representative velocities
of 0.01 m/s, 0.02 m/s, and 0.04 m/s were selected. At the velocity of 0.01 m/s, the
displacement interface is relatively smooth, showing a relatively stable advancing pat-
tern of the displacement front. When the velocity increases, the displacement interface
becomes more disordered, and the distribution of the air volume fraction becomes
more scattered. This phenomenon arises from a fundamental shift in hydrodynamic
dominance: at lower velocities, viscous forces govern flow behavior, while increasing
velocity progressively amplifies inertial effects. The dynamic interplay between these
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competing forces dictates the evolution of interfacial morphology, sweep coverage,
and displacement efficiency. Moreover, under low-velocity conditions, while sweep
efficiency φ reaches its peak value, the sluggish advancement of the displacement
front results in diminished volumetric displacement rates per unit time, leading to
suboptimal displacement efficiency on a temporal basis. When the velocity increases,
the sweep coverage expands; however, due to excessive viscous fingering, this results
in a non-uniform displacement process, lower sweep efficiency, and a higher residual
volume of non-Newtonian fluid. Therefore, in various engineering applications, both
sweep efficiency and displacement efficiency should be comprehensively considered
to determine (or achieve) an appropriate overall recovery efficiency.

(3) In the variant study of the classical HSC model, four depth gradients, −0.5◦, −1◦,
+0.5◦ and +1◦, were introduced. It was found that both the direction and magnitude
of the depth gradient significantly influence the morphology and extension length of
viscous fingering. Both positive and negative depth gradients promote the develop-
ment of fingering on one side. In the absence of a gradient, the two fingers develop
relatively uniformly. This phenomenon fundamentally stems from flow field redistri-
bution induced by depth variations along the flow conduit. The presence of a depth
gradient establishes preferential flow pathways in specific spatial orientations, thereby
amplifying fingering instabilities through localized flow intensification. Furthermore,
as the depth gradient changes from negative to positive, that is from −1◦ to +1◦,
the maximum velocity within the fingers gradually decreases. This occurs because
the increasing gradient magnitude transforms the flow channel from a narrowing
configuration to a widening one, thereby causing a reduction in velocity. Therefore,
depth gradients fundamentally modulate the morphology and propagation of viscous
fingering by dynamically altering the geometric confinement of flow conduits, thereby
strategically reconfiguring the spatial distribution of velocity fields. In engineering
applications, deliberate modulation of depth gradients within flow conduits enables
strategic control over displacement dynamics, facilitating targeted suppression or
triggering of interfacial instabilities as required by operational objectives.
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